Author’s response to reviews

Title: An Investigation of the Effects of Infertility on Women's Quality of Life: A Case-Control Study

Authors:

Katayoun Bakhtiar (k_bakhteyar@yahoo.com)
Ramin Beiranvand (beiranvand83@yahoo.com)
Arash Ardalan (arashard@gmail.com)
Farahnaz Changaee (farahnazchangavi@yahoo.com)
Mohammad Almasian (almasian26@gmail.com)
Afsaneh Badrizadeh (badrizadeha@yahoo.com)
fatemeh bastami (fatemeh2011bastami@gmail.com)
Farzad Ebrahimzadeh (farzadebrahimzadeh@gmail.com)

Version: 2 Date: 21 May 2019

Author’s response to reviews:

BMC Women's Health

Dear Editor

We greatly appreciate your encouraging reply and the reviewers’ constructive and positive comments on our original manuscript entitled "An Investigation of the Effects of Infertility on Women’s Quality of Life: A Case-Control Study". Based on the comments of Editor, we have revised our manuscript. The changes in the newly revised manuscript are highlighted in red and blue.
We would like to re-submit our newly revised manuscript to the BMC Women's Health, and hope that it is acceptable for publication in your journal. We look forward to hearing from you soon.

Yours sincerely,

Editor Comments:

1. Page 7 line 4 '. women (20%) than fertile ones (10.4%) (p = 0.001 ). Table 2..." The p-value should be 0.023 instead of 0.001 according to table 1.

   The p-value was revised.

2. please check the grammar on second paragraph on page 7

   the grammar was checked and be revised. revisions are highlighted in red and blue.

3. Please check the grammar and p-values on third paragraph on page 7.

   the grammar was checked and be revised. The p-values were revised on third paragraph on page 7. revisions are highlighted in red and blue

4. Page 7 line 39-40. The statement is not true according to table 2.

   The table 2 was revised.

5. Please correct the p-values on lines 54,56, and 59 (all three p-values).

   the p-values were corrected on lines 54,56, and 59 (all three p-values).
6. Please delete the "?" at end of the paragraph on line 59
"?" was deleted.

7. Please correct the p-value on line 26, page 8. I should be 0.046 instead of 0.064.
the p-values were corrected on lines line 26, page 8.

8. Please correct the numbers in table for Type of infertility treatment, many values of mean and sd were switched.
The numbers were revised in table for Type of infertility treatment

8.1 for physical health dimension column, the numbers for IUI/ICSI were apparently switched
the numbers The numbers were revised in table for for IUI/ICSI

8.2 for psychological health dimension column, all the numbers except for IVF were probably switched.
All the numbers were revised

8.3 for environmental health dimension column, same errors as for 8.2 above.
It was revised.

8.4 for social health dimension column, same errors as for 8.2 above.
It was revised.

8.4 for total score of quality of life, same errors as for 8.2 above.
It was revised.